Author's response to reviews

Title: Southampton Mealtime Assistance Study: design and methods

Authors:

Helen C Roberts (hcr@soton.ac.uk)
Anna L Pilgrim (alp@mrc.soton.ac.uk)
Marinos Elia (M.Elia@soton.ac.uk)
Alan A Jackson (A.A.Jackson@soton.ac.uk)
Cyrus Cooper (cc@mrc.soton.ac.uk)
Avan Aihie Sayer (aas@mrc.soton.ac.uk)
Sian M Robinson (smr@mrc.soton.ac.uk)

Version: 3 Date: 22 August 2012

Author's response to reviews:

Academic Geriatric Medicine
University of Southampton
Southampton General Hospital Mailpoint 807
Southampton
SO16 6YD
UK
17th August 2012

Dear Colleague
The following comments from Dr Crow have been addressed in this revision:
‘I would also like to remind you to please revise your manuscript to include the trial registration number as the final line in the abstract, and to include a statement of ethical approval.’

The following revisions have been made to the original manuscript:
1. The following sentence has been added at the end of the abstract;
‘Trial registered with Clinicaltrials.gov NCTO1647204.’

2. The final sentence of the section on study design within the methods section now reads;
‘The study received full approval by the local research ethics committee.’

Please let me know if I can be of any further help.

Best wishes
Helen Roberts